Author's response to reviews

Title: Non-Invasive Respiratory Volume Monitoring Identifies Opioid-Induced Respiratory Depression in an Orthopedic Surgery Patient with Diagnosed Obstructive Sleep Apnea: A Case Report

Authors:

Christopher Voscopoulos (christopher.voscopoulos@gmail.com)
Eamon Fleming (eamon.fleming@respiratorymotion.com)
Edward George (edgeorge@partners.org)

Version: 2
Date: 3 December 2014

Author's response to reviews:

Hello, the following revisions were made to the manuscript:

The CARE checklist was typed for easier review
The abstract was restructured into 3 sections
The ethnicity of the patient was added to the case description
A list of abbreviations was added to the end of the manuscript (the abbreviations are also in the text directly following the first use of the term